Author's response to reviews

Title: An uncommon reason for failure of CT colonoscopy: a case report

Authors:

Angeliki Neroladaki (angeliki.Neroladaki@hcuge.ch)
Romain Breguet (romain.breguet@hcuge.ch)
Diomidis Botsikas (diomidis.botsikas@hcuge.ch)
Sylvain Terraz (sylvain.terraz@hcuge.ch)
Christoph D Becker (christoph.becker@hcuge.ch)
Xavier Montet (xavier.montet@hcuge.ch)

Version: 2 Date: 7 October 2011

Author's response to reviews:

In this revised manuscript, we have made all the changes asked. More specifically:

1. We have now indicated in the title that this is a case report.
2. The abstract now start at page 2 and has the following section headings: Introduction, case presentation and conclusion
3. We have added the patient ethnicity
4. We have added a conclusion section at the end of the manuscript
5. We have added a Patient Consent section
6. WE have added a Competing interests section
7. We have added a Author’s Contribution section
8. The Title page have been revised accordingly to your comment

Thank you for your time on our behalf.

Sincerely,

Xavier Montet